Author's response to reviews

Title: Nevirapine versus Efavirenz-based highly active antiretroviral therapy regimens in antiretroviral-naive patients with HIV and Tuberculosis infection in India: A Randomized Control Trial.

Authors:

Sanjeev Sinha (drsanjeevsinha2002@yahoo.com)
P Raghunandan (raghunandan88@gmail.com)
Rahul Chandrashekhar (rahulcshekhar@gmail.com)
Surendra K Sharma (sksharma.aiims@gmail.com)
Sanjiv Kumar (sanjeevchaudhary.aiims@gmail.com)
Sahajal Dhooria (sahajal@gmail.com)
Meera Ekka (drmeera_2004@yahoo.co.in)
Velpandian T (tvelpandian@yahoo.com)
Sanjay Ranjan (drs.r@rediffmail.com)
Hafeez Ahmad (hafeezaiims@gmail.com)
J C Samantaray (jsamantaray@yahoo.com)
S Venkatesh (drsvenkatesh1@gmail.com)
B B Rewari (drbbrewari@gmail.com)
R M Pandey (rmpandey@gmail.com)

Version: 2 Date: 26 March 2013

Author's response to reviews: see over
Dear Sir,

Please find enclosed the manuscript: “Nevirapine versus Efavirenz-based highly active antiretroviral therapy regimens in antiretroviral-naive patients with HIV and Tuberculosis infection in India: A Randomized Control Trial”, by Sanjeev Sinha, et al., submitted as a new article to *BMC Infectious Diseases* for consideration of publication. All co-authors have seen and agree with the contents of the manuscript, and there is no conflict of interest to report. We certify that the submission is an original work and is not under review at any other publication.

This manuscript presents the findings of a randomized controlled clinical trial conducted at All India Institute of Medical Sciences, New Delhi, India. This study could be of considerable interest to the readership of *BMC Infectious Diseases* because it brings new light in management of TB/HIV co-infection, suggesting a shift from the prevalent practice.

We hope that the editorial board and the reviewers will agree on the interest of this study. Looking forward to a favourable response.

Sincerely yours,

Sanjeev Sinha, on behalf of the authors.

Corresponding author:

Dr Sanjeev Sinha,
Additional Professor,
Department of Medicine,
All India Institute of Medical Sciences,
Ansari Nagar, New Delhi- 110029, India.
Email: drsanjeevsinha2002@yahoo.com,
phone number: +91-11-64503006, 26594440,
Fax number: +91-11-26588866.